Author's response to reviews

Title: Evaluation of HIV protease and nucleoside reverse transcriptase inhibitors on proliferation, necrosis, apoptosis in intestinal epithelial cells and electrolyte and water transport and epithelial barrier function in mice

Authors:

Manuel B Braga Neto (mbn8w@virginia.edu)
Carolina V Aguiar (lalaiveras@hotmail.com)
Jamilly G Maciel (jamillygomes@walla.com)
Bruna M Oliveira (castrobruna@hotmail.com)
Jesus E Sevilleja (jes3gb@virginia.edu)
Reinaldo B Oria (oria@ufc.br)
Gerly A C Brito (gerlybrito@hotmail.com)
Cirle A Warren (ca6t@virginia.edu)
Richard L Guerrant (rlg9a@virginia.edu)
Aldo A M Lima (alima@ufc.br)

Version: 8 Date: 11 August 2010

Author's response to reviews:

Dear Editorial Team,

We have made minor modifications in the main manuscript, which have been highlighted accordingly. Additionally, we have also made minor modifications in Figures 1, 3, 4 and 8.

Sincerely,

Manuel Braga Neto